Author's response to reviews

Title: Seroprevalence of transfusion-transmissible infections and evaluation of the pre-donation screening performance at the Provincial Hospital of Tete, Mozambique

Authors:

Jocelijn Stokx (stokxjocelijn@hotmail.com)
Philippe Gillet (pgillet@itg.be)
Anja De Weggheleire (Adeweggheleire@itg.be)
Esther C Casas (estherccasas@gmail.com)
Rosa Maendaenda (romaenda@hotmail.com)
Adelino J Beulane (abeulane@hotmail.com)
Ilhes Jani (ijani@email.com)
Solon Kidane (kida0004@yahoo.com)
Carla D Mosse (carladasdoresmosse@yahoo.com.br)
Jan Jacobs (jjacobs@itg.be)
Emmanuel Bottieau (EBottieau@itg.be)

Version: 4 Date: 18 May 2011

Author's response to reviews:

Dear Editor,

Please find enclosed our final manuscript "Seroprevalence of transfusion-transmissible infections and evaluation of the pre-donation screening performance at the Provincial Hospital of Tete, Mozambique" with all requested changes.

Please find in this cover letter a point-by-point response to your final requests.

The order of the authors have been reorganized as it should be (there was probably a mistake at the first submission). Also the style of authors’ names fits now with your requirements.

All highlighting and tracking of the previous revised version have been removed (changes accepted).

The vertical lines of the Table 2 have been removed.

The whole document has been rechecked once again for typographical errors.

Yours sincerely,

Emmanuel Bottieau
ebottieau@itg.be